Author's response to reviews

Title: Cauda Equina Compression in Achondroplastic Dwarf. Is complex anterior and posterior surgical intervention necessary?

Authors:

George S SAPKAS (gsapkas1@hol.gr)
Konstantinos KATEROS (kkateros@hotmail.com)
Stamatios A PAPADAKIS (snapmd@gmail.com)
George A MACHAIRAS (gmacheras@gmail.com)
Emmanuel Brilakis (brilakis@mycosmos.gr)

Version: 2 Date: 10 August 2008

Author's response to reviews:

Dear Sirs,

We dispatch you for consideration the manuscript titled “Cauda Equina Compression in Achondroplastic Dwarf. Is complex anterior and posterior surgical intervention necessary?”. This manuscript describes a rare case of an achondroplastic dwarf who presented with partial paraplegia due to cauda equina compression. We describe a different treatment from the one recommended by the literature due to the presence of kyphosis. We believe that this study is relevant to the scientific field that your journal covers, and we are looking forward for your review.

With Regards
George S. SAPKAS
Konstantinos KATEROS
Stamatios A. PAPADAKIS
George A. MACHAIRAS
Emmanuel BRILAKIS